Author’s response to reviews

Title: PREVALENCE OF GOITER AND ITS ASSOCIATED FACTORS AMONG PRIMARY SCHOOL CHILDREN IN CHOLE DISTRICT, ARSI ZONE, ETHIOPIA: A CROSS SECTIONAL STUDY

Authors:
Takele Adilo (admetakele@gmail.com)
Abera Bekele (aberabekele@gmail.com)

Version: 2 Date: 14 Sep 2018

Author’s response to reviews:

To: Editor Office,
BMC Nutrition Journal

Subject: Submitting the second revised version of manuscript to be published in your reputable Journal, BMC Nutrition

Dear Editorial Office Manger,

We are pleased to inform you kindly that the third version of our submitted manuscript to be published/ revised for the second time/ is attached here with. The given comments were well taken and all possible attempts to address them one by one adequately were made. The title of the manuscript is “PREVALENCE OF GOITER AND ITS ASSOCIATED FACTORS AMONG PRIMARY SCHOOL CHILDREN IN CHOLE DISTRICT, ARSI ZONE, ETHIOPIA: NUTN-D-18-00032”

In general, we want to say thank you very much once more for your very positive and helpful comments and/or queries so far given for improving the manuscript and hope to get our research article published in your reputable journal

Looking forward to hearing your fast and positive response as usual,
Sincerely,

Takele Menna and Abera Bekele

Corresponding author: Takele Menna, PhD

Department of Public Health, SPHMMC, Addis Ababa, Ethiopia,

E-mail address: admetakele@gmail.com; Cell Phone: +251911134934